Author’s response to reviews

Title: Evaluation of Lifestyle Interventions to Treat Elevated Cardiometabolic Risk in Primary Care (E-LITE): A Randomized Controlled Trial

Authors:

Jun Ma (maj@pamfri.org)
Abby C King (king@stanford.edu)
Sandra R Wilson (wilsongs@pamfri.org)
Lan Xiao (xiaol@pamfri.org)
Randall S Stafford (rstafford@stanford.edu)

Version: 2 Date: 2 October 2009

Author’s response to reviews: see over
September 24, 2009

Melissa Norton, MD
Editor-in-Chief, BMC series
34-42 Cleveland Street, London, W1T 4LB, UK

Re: New Manuscript Submission

Dear Dr. Melissa Norton:

Please find enclosed a manuscript entitled “Evaluation of Lifestyle Interventions to Treat Elevated Cardiometabolic Risk in Primary Care (E-LITE): A Randomized Controlled Trial” to be considered for publication as a Study Protocol. The study is funded by the National Institute of Diabetes and Digestive and Kidney Diseases (NIDDK) and the American Heart Association (AHA). The manuscript is being submitted only to BMC Family Practice and will not be submitted elsewhere while under consideration. Should it be published in the journal, it will not be published elsewhere without permission of the editors. The authors have each participated sufficiently in the work to take public responsibility for appropriate portions of the content. The authors declare that they have no financial, research, organizational, or other interests to disclose that are relevant to the execution of this research or this publication. All authors have seen and approved the manuscript, as submitted.

Thank you for your careful consideration!

Sincerely,

Jun Ma, M.D., Ph.D.